Author's response to reviews

Title: Analysis of awareness of health knowledge among rural residents in western China

Authors:

Fang Yuan (yuanfanglanzhou@163.com)
Dongfu Qian (Qian-qiandf006@hotmail.com)
Chenglong Huang (righthuang@sina.com)
Miaomiao Tian (shuishuipp@163.com)
Yuanxi Xiang (11427843@qq.com)
Zhanchun Feng (zcfeng@hust.edu.cn)

Version: 3 Date: 5 May 2014

Author’s response to reviews:

Dear Editor,

Here within enclosed is our paper for consideration to be published on “BMC Public Health”. The further information about the paper is in the following:

The title: Analysis of awareness of health knowledge among rural residents in west China

Information of authors:

Fang Yuan1, Dongfu Qian2 Chenglong Huang3, Miaomiao Tian4, Yuanxi Xiang1, Zhanchun Feng*1

1 School of Medicine and Health Management, Tongji Medical College, HuaZhong University of Science and Technology, Wuhan, HuBei Province, China. Email: Fang Yuan- yuanfanglanzhou@163.com; Yuanxi Xiang-11427843@qq.com; Zhanchun Feng* - zcfeng@hust. edu.cn

2 College of Medical Administration, Nanjing Medical University, Nanjing, Jiangsu Province, China. Email: Dongfu Qian-Qian-qiandf006@hotmail.com

3 College of Laboratory Medicine, Chongqing Medical University, Chongqing, Chongqing, China Email: Chenglong Huang – righthuang@sina.com

4 Institute of Medical Information, Center for Health Policy and Management, Chinese Academy of Medical Sciences, Beijing, China. Email: Miaomiao Tian – shuishuipp@163.com

* Corresponding author
1. All authors of this research paper have directly participated in the planning, execution, or analysis of this study.
2. All authors of this paper have read and approved the final version submitted.
3. The authors claim that none of the material in the paper has been published or is under consideration for publication elsewhere.
4. The contents of this manuscript will not be copyrighted, submitted elsewhere, while acceptance by your journal.
5. There are no directly related manuscripts or abstracts, published or unpublished, by any authors of this paper.

We believe the paper may be of particular interest to the readers of your journal because the study is in order to explore how to evaluate health needs of rural residents reasonably, formulate and improve the health promotion activities of rural residents, so as to improve the health level of residents.

Correspondence and phone calls about the paper should be directed to Prof. Zhanchun Feng at the following address and E-mail address:

Zhanchun Feng  
School of Medicine and Health Management  
Tongji Medical College of Huazhong University of Science & Technology  
Wuhan, PRC  
Email: zcfeng@hust.edu.cn

Thanks very much for your attention to our paper.

Sincerely yours,

Zhanchun Feng